Author's response to reviews

Title: Rectal gel application of Withania somnifera root extract denotes anti-inflammatory and muco-restorative activity in TNBS-induced Inflammatory Bowel Disease

Authors:

Pankaj Pawar (pnkjpawar@yahoo.com)
Suhit Gilda (suhit.gilda@gmail.com)
Siddhesh Sharma (sid_vns@yahoo.com)
Suresh Jagtap (chiritatml@rediffmail.com)
Anant Paradkar (arparadkar@gmail.com)
Kakasaheb Mahadik (krmahadik@rediffmail.com)
Prabhakar Ranjekar (ranjekar@vsnl.net)
Abhay Harsulkar (aharsulkar@yahoo.com)

Version: 3 Date: 12 April 2011

April 12, 2011
To,
Editor,
BMC Complementary & Alternative Medicine

Dear Sir,

Please find attached herewith a Proof corrected manuscript entitled, “Rectal gel application of Withania somnifera root extract denotes anti-inflammatory and muco-restorative activity in TNBS-induced Inflammatory Bowel Disease” for the publication.

Following are the Point-by-point description of the changes made

Major revisions (we require the author to make these changes)

Title: please ensure that the title is identical in the manuscript and the submission system. Currently they do not match.

Change: The title is now identical in the MS and in the submission system

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). Currently they do not match.

Change: The author list is now identical with that of the MS

Abstract: please ensure that the Abstract is present in the manuscript and
matches exactly what is in the submission system. Currently they do not match.

Change: The abstract in MS and in the submission system is now identical

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.

Change: The Tables are now included in the Manuscript after the reference section

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the country in the affiliation details.

Change: The country name is now added

Please remove any blank pages from the manuscript.

Change: There are no blank pages in the MS

Materials and methods: Please rename this section 'Methods'.

Change: Materials and methods now reads are Methods

Figures: Please ensure all mention of figures is consistent (it should be Figure or Fig but not a mixture of both).

Change: It is now consistent and reads as Fig everywhere in the MS

Editorial note: Reference no 4: Authors need to change 'Pottent lipid peroxidation inhibitors...' to 'Potent lipid peroxidation inhibitors....'

Change: The correction has been made

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Change: The Ms is now reviewed for the typos and rectified

Thanking you and with warm regards.

Sincerely,

Prof. Abhay Harsulkar